Author’s response to reviews

Title: Nurse-led Home Visitation Programme to Improve Health-Related Quality of Life and Reduce Disability Among Potentially Frail Community-Dwelling Older People in General Practice: A Theory-Based Process Evaluation

Authors:

Mandy MN Stijnen (mandy.stijnen@maastrichtuniversity.nl)
Maria WJ Jansen (maria.jansen@ggdzl.nl)
Inge GP Duimel-Peeters (inge.duimel@maastrichtuniversity.nl)
Hubertus JM Vrijhoef (H.J.M.Vrijhoef@uvt.nl)

Version: 2 Date: 19 March 2014

Author’s response to reviews:

Please find enclosed our manuscript, now including line and page numbers as requested. Furthermore, we included a statement in the manuscript of ethical approval and informed consent obtained from participants.